

Earth's Future

RESEARCH ARTICLE

10.1029/2021EF002600

Key Points:

- Wave damage to coral communities from tropical cyclones depends on their intensity, size and duration while tracking near coral reefs
- Downscaled tracks capture observed cyclone characteristics impacting the Great Barrier Reef with mixed to poor performance in other regions
- Projections vary by region and cyclone characteristic with some models projecting increases and others decreases in future reef damage

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. M. Dixon, ee17amd@leeds.ac.uk

Citation:

Dixon, A. M., Puotinen, M., Ramsay, H. A., & Beger, M. (2022). Coral reef exposure to damaging tropical cyclone waves in a warming climate. *Earth's Future*, *10*, e2021EF002600. https://doi. org/10.1029/2021EF002600

Received 10 DEC 2021 Accepted 30 JUN 2022

Author Contributions:

Conceptualization: Adele M. Dixon, Marji Puotinen Formal analysis: Adele M. Dixon Methodology: Adele M. Dixon, Marji Puotinen, Hamish A. Ramsay Project Administration: Adele M. Dixon Resources: Marji Puotinen Supervision: Maria Beger Validation: Adele M. Dixon Visualization: Adele M. Dixon Writing – original draft: Adele M. Dixon

© 2022 The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Coral Reef Exposure to Damaging Tropical Cyclone Waves in a Warming Climate

Adele M. Dixon^{1,2}, Marji Puotinen³, Hamish A. Ramsay⁴, and Maria Beger^{1,5}

¹School of Biology, Faculty of Biological Sciences, University of Leeds, Leeds, UK, ²Priestley International Centre for Climate, University of Leeds, Leeds, UK, ³Australian Institute of Marine Science, Crawley, WA, Australia, ⁴CSIRO, Oceans and Atmosphere, Aspendale, VIC, Australia, ⁵Centre for Biodiversity and Conservation Science, School of Biological Sciences, University of Queensland, Brisbane, QLD, Australia

Abstract Tropical cyclones generate large waves that physically damage coral communities and are commonly cited as a worsening threat to coral reefs under climate change. However, beyond projections of ocean basin-scale changes in cyclone intensity, the other determinants of future coral reef damage such as cyclone size and duration remain uncertain. Here, we determine the extent to which downscaled cyclones represent observed cyclone characteristics that influence wave damage to Australian coral reef regions. We then investigate mid-century (2040-2060) and end of century (2080-2100) downscaled tracks to assess whether cyclone characteristics will change with future warming under a high-emissions scenario. We find that spatial uncertainties in downscaled cyclogenesis and track positions limit estimates of reef damage for individual coral reefs and regions. Further, the models are unable to reproduce the most reef damaging cyclones for any of the regions. The downscaled tracks capture observed cyclone characteristics, such as size, impacting the Great Barrier Reef well, but perform poorly for the Northern Territory, with mixed performance for the Coral Sea and Western Australia. We find no clear evidence that cyclones will cause more damage to Australian coral reef regions in the future, at least based on the climate models and downscaling approach examined here. There is increasing interest in using tropical cyclone projections to assess future coral reef exposure to damaging waves. We recommend caution when interpreting such projections due to large uncertainty in the mechanisms that influence reef damaging tropical cyclone characteristics and how these will change with future warming.

Plain Language Summary Tropical cyclone intensity, size and duration together determine the extent to which their waves damage coral reefs. Increased tropical cyclone intensity with climate change is often cited as evidence that tropical cyclones will cause more damage to coral reefs in the future but changes to size and duration remain uncertain. Here, we determine whether tropical cyclones simulated from climate models can represent the observed tropical cyclone characteristics that are important for estimating wave damage to coral reefs and assess how these characteristics might change in the future. We find that the tropical cyclones simulated from climate models capture the observed cyclone characteristics well for those impacting the Great Barrier Reef (with the exception of the most damaging cyclones) with mixed to poor performance for other regions. The projections of future reef damage are uncertain with some models projecting increases and others decreases. Tropical cyclone projections are used in conservation planning to identify and protect the coral reefs least exposed to future tropical cyclones. However, we find that the simulated tropical cyclone tracks explored here are unlikely to represent future reef damaging tropical cyclone characteristics well if used in conservation decision making.

1. Introduction

Tropical cyclones are often a key driver of coral reef condition where they commonly occur (De'ath et al., 2012; Zinke et al., 2018), potentially thwarting conservation success if not considered in management decisions. For example, protecting sites least exposed to thermal stress and able to provide larval recruits to other vulnerable reefs as a strategy for enhancing coral reef survival (Beyer et al., 2018) will fail if such sites are subjected to frequent and severe wave damage from tropical cyclones. The threat of tropical cyclones is rising as the increased severity and spatial extent of other stressors impedes recovery from storm damage (Blackwood et al., 2011; Dietzel et al., 2021). Conservation plans need to consider multiple environmental dimensions of climate change (Dixon et al., 2021; Groves et al., 2012) because climate change can lead to ecosystem collapse (Newton et al., 2021). For coral reefs, a key part of this is to identify the management areas experiencing the most damaging tropical



Writing – review & editing: Adele M. Dixon, Marji Puotinen, Hamish A. Ramsay, Maria Beger cyclones now and in the future. Thus, there is increasing interest in the use of future simulated tracks in coral reef vulnerability assessments and conservation planning, though there has been no assessment of their suitability nor their ability to project robust changes in reef damage at the coral reef region scale. Here, we examined to what degree simulated historical cyclone tracks likely capture the key characteristics that underpin a cyclone's ability to damage reefs, assuming vulnerable colonies are present. Secondly, we determined whether robust changes in reef damaging tropical cyclone characteristics are projected at the scale of coral reef regions in tropical Australia.

Despite being localized short-term individual events, repeated tropical cyclones can cause long-term damage to coral reefs across broad scales. Recovery from this can take decades to centuries (Harmelin-Vivien, 1994), especially if damage affects the physical structure of the reef (Hughes & Connell, 1999) or recovery is impeded by other disturbances (Hughes, 1994). Yet, spatial variability in coral reef exposure to tropical cyclone waves at both local and regional scales means that damage is patchy (Beyer et al., 2018; Gilmour et al., 2019; Maynard et al., 2016; Puotinen et al., 2016; Wolff et al., 2016; Zinke et al., 2018), not least because equatorial coral reefs are outside the geographic range where tropical cyclones track (Puotinen et al., 2020). Tropical cyclones degrade coral reef ecosystems in various ways (Harmelin-Vivien, 1994). Heavy rainfall and flooding lowers salinity and increases nutrient concentration and terrestrial sediment influx, and large waves cause sediment resuspension and physical damage. Physical damage to coral communities ranges from breakage of branches in the most vulnerable and delicate species to the removal of entire sections of the reef structure (Beeden et al., 2015; Harmelin-Vivien, 1994; Puotinen et al., 2016).

The severity of physical damage from waves is dependent on storm characteristics such as intensity, size, duration and translation speed (Puotinen et al., 2020) as well as local-scale reef characteristics such as depth, structural complexity, community type and disturbance history (Blackwood et al., 2011; Harmelin-Vivien, 1994). At the coral colony scale, damage is always patchy because the individual characteristics (morphology, size) and spatial arrangement of colonies determine the extent to which physical damage from a potentially damaging wave climate actually occurs (Madin & Connolly, 2006). The worst possible wave damage can be expected from cyclones that are intense, large, and slow-moving with long-lived tracks that persist near reefs. This damage is most fully realized for vulnerable colonies within reef communities that are compromised by other stressors or are not routinely exposed to a high energy wave regime (Madin & Connolly, 2006). Tropical cyclone-induced wave damage can be exacerbated by other climate stressors, for example, where ocean acidification weakens coral skeletons and thus increases their vulnerability to physical damage (Madin et al., 2012).

Global climate models project increases in the frequency and severity of thermal stress events in coral reef regions in the future (Dixon et al., 2022). Though tropical cyclone projections have existed for coral reef regions for many years, they have focused on global and ocean basin-scale trends in tropical cyclone frequency and intensity (Emanuel et al., 2008; Knutson et al., 2013). At the global scale, there is medium to high confidence that the average tropical cyclone peak intensity and the proportion of storms that reach high intensity will increase in the future with climate change, but trends in the remaining tropical cyclone characteristics that influence wave-induced reef damage are less certain (Knutson et al., 2020). Future changes to translation speed are equivocal; one study has projected a significant decrease (Gutmann et al., 2018), while others project no change (Kim et al., 2014; Knutson et al., 2013; Wu et al., 2014). Projected changes in tropical cyclone size are also highly variable between studies: some studies project increases in some basins (Kim et al., 2014; Knutson et al., 2015; Yamada et al., 2017), and decreases in others (Knutson et al., 2015; Yamada et al., 2017), while other studies project no change (Gutmann et al., 2018).

The projected global and basin-scale increases in tropical cyclone intensity are often cited as evidence of greater tropical cyclone-induced coral reef damage with future climate change (Cheal et al., 2017; França et al., 2020; Gilmour et al., 2019; Harvey et al., 2018), but there are two key gaps in knowledge undermining this statement. First, global and basin-scale changes in tropical cyclone characteristics do not capture spatial variation in future tropical cyclone exposure within and between coral reef regions. For example, whether the projected increase in intensity differs between the coral reef regions within an ocean basin is unknown, much less between individual reefs. Second, a projected increase in intensity alone does not provide a complete picture of the future potential for coral reef damage as size, duration and location of tropical cyclone tracks also determine the coral reef wave damage severity and extent. For example, tropical cyclones may be more intense in the future but no longer track as close to coral reefs, be as large in size or move as slowly.





Figure 1. Map of Australia showing 100 km boundaries around the Coral Sea, Great Barrier Reef and Western Australia coral reef areas and the entire Northern Territory region. Coral reef areas are shown in black. The bottom panel shows a track position with its spatial footprint (maximum radius to gales) intersecting with the Western Australia region.

Vital for assessing damage potential to coral reefs is understanding where cyclones are likely to form and track. Numerous factors influence tropical cyclone formation including sea surface temperature, vertical wind shear, mid-level moisture and the Coriolis parameter (Emanuel, 2003). The link between tropical cyclone formation and climate is not fully understood, making projected changes in tropical cyclone formation with future climate change uncertain (Sobel et al., 2021; Walsh et al., 2016). Where a tropical cyclone tracks, and how quickly, is influenced by synoptic-scale atmospheric circulations, including major subtropical high pressure systems (Camp et al., 2019; Chan & Gray, 1982; Chu et al., 2012). Where a tropical cyclone tracks can affect its intensity and lifetime, as high sea surface temperature can cause a cyclone to intensify while low sea surface temperature, high wind shear and landfall can cause tropical cyclones to weaken and dissipate (Emanuel, 2003). Hence, the ability of climate models to project future changes in tropical cyclone characteristics hinges on the accurate simulation of the many mechanisms affecting the tropical cyclone life cycle.

Future changes to tropical cyclone characteristics are commonly estimated at global or ocean basin (e.g., South Pacific Ocean) scales. Only once have any projections of future changes for individual coral reef regions been undertaken (Great Barrier Reef, where Callaghan et al. (2020) used the HADGEM model to assess the extent to which reefs prevent coastal erosion from cyclones). We build upon that work to assess the current and future potential for cyclone waves to damage reefs based on a suite of six CMIP5 climate models and extending the study region to all of tropical northern Australia. Ideally, a numerical wave model would be used to simulate near-shore wave climates for future cyclones using high resolution bathymetry as per Callaghan et al. (2020) for the full suite of climate models and for all regions. However, running numerical wave analysis for large sets (thousands) of historical and future simulated tracks is computationally and time intensive, the requisite bathymetry data does not yet exist for Western Australia nor the Northern Territory, and model performance for the full suite of climate

models has not yet been assessed. We therefore examined whether downscaled tropical cyclones (Emanuel et al., 2006, 2008) derived from historical climate model output can capture observed regional variability in individual potential reef damaging tropical cyclone characteristics such as intensity, size and duration near reefs. We used tropical Australia as a case study, as its northern coastlines are frequently exposed to tropical cyclone impacts (Chand et al., 2019) and there is regional variability in coral reef exposure to cyclone generated waves (Great Barrier Reef—Maynard et al., 2016; Puotinen et al., 2016; Western Australia—Gilmour et al., 2019; all tropical Australia—Puotinen et al., 2020). We then compared regional tropical cyclone reef damaging characteristics between the past and future climates under the high-emissions Representative Concentration Pathway (RCP) 8.5 scenario.

2. Materials and Methods

2.1. Calculating Observed Reef Damaging Tropical Cyclone Metrics

We identified tropical cyclones with the potential to have generated waves capable of damaging tropical coral reefs, henceforth called "reef damaging" cyclones, as those where the cyclone circulation (defined as maximum radius to gale force winds (~17 m/s) in any sector mapped around the cyclone track position each hour) intersects a 100 km boundary around any of three regional coral reef regions (Coral Sea, Great Barrier Reef and Western Australia; Figure 1)—termed the track position's spatial "footprint." This extends work done previously for the Great Barrier Reef (Wolff et al., 2018) by explicitly mapping how the size of the cyclone (the spatial footprint) varies along the track and intersecting that with the coral reef regions instead of assuming damage is possible at any reef within a uniform zone around each cyclone track. The 100 km zone was added to conservatively account for positional uncertainty in recorded cyclone track positions. In the satellite era (post 1970), such uncertainty is

less than 150 km, and less than 50 km where the tropical cyclone has a clearly defined eye (D. Herndon, personal communication). We included a fourth region, Northern Territory, but due to uncertainty regarding the location of coral reefs in the Northern Territory we used the entire region rather than a 100 km buffer around the reef area (Figure 1).

Though cyclones can generate freshwater flood plumes whose lowered salinity and light-blocking turbidity can adversely affect coral communities (Brodie et al., 2012), here we limited our consideration to physical damage to corals from cyclone generated waves. As the magnitude of a given sea state depends on the combination of wind speed, duration and fetch (distance over which winds can blow consistently unobstructed), three key characteristics of cyclones contribute most to their potential to generate seas capable of damaging reefs: intensity, size of circulation, and duration near reefs (Puotinen et al., 2020). For each incidence of a tropical cyclone's spatial footprint intersecting a coral reef region between 1985 and 2020, the following three reef damaging tropical cyclone characteristics were calculated: maximum intensity, maximum radius to gales and duration of gales. Best track data prior to the early 1980s has a high number of missing records and tends to underestimate intensity (Ramsay et al., 2012). We further applied a reef damage index (*RDI*) that uses all three characteristics to estimate damage potential based on field data of past cyclone damage of reefs predominantly in the Great Barrier Reef (Table S1 in Supporting Information S1).

The three metrics were calculated only for the cyclone track positions located along the section of tracks whose spatial footprint intersects with the reef regions. The maximum intensity was calculated using the maximum 10-min wind speed extracted from the International Best Track Archive for Climate Stewardship (IBTrACS) (Knapp et al., 2010, 2018). The maximum radius to gales was calculated using the maximum distance from the cyclone eye to gale force winds at 17 m/s in any of the four geographic quadrants, reported in IBTrACS. Where the radius to gales was missing in IBTrACS, the maximum radius to gales from the Bureau of Meteorology database was used. Where the radius to gales was missing in both data sources, the basin average (210 km) was used (Chavas & Emanuel, 2010). Using a basin rather than a regional average avoids the modifiable areal unit problem where the average radius to gales may change depending on where the regional boundaries are drawn. The duration of gales is the number of days a tropical cyclone's spatial footprint intersects with a reef region while the maximum wind speed is greater than 17 m/s.

The reef damage index (*RDI*) was used to estimate the overall potential for reef damage from waves (assuming vulnerable corals are present) for each tropical cyclone track while impacting a region. The index is calculated by:

$$RDI = \left(\frac{MI}{MI_{\text{max}}}\right) * \left(\frac{RG}{RG_{\text{max}}}\right) * \left(\frac{DG}{DG_{\text{max}}}\right) * 100$$

where *MI* is the maximum intensity, *RG* is the maximum radius to gales and *DG* is the duration of gales. The reef damage index is a reasonable estimate of the reef damage potential of a tropical cyclone when compared to field data for nine cyclones from the Great Barrier Reef and Western Australia (Table S1 in Supporting Information S1).

The greatest potential for wave damage comes from a cyclone that is simultaneously intense, slow moving, large and generates gales for a long time within a reef region. No recorded cyclone in any of the Australian coral reef provinces has met all of these criteria. Cyclone Yasi in 2011 on the Great Barrier Reef (Beeden et al., 2015) and Cyclone Lua in 2012 in Western Australia (Puotinen et al., 2020) were simultaneously both intense and large, but not slow-moving. Cyclone Hamish in 2009 on the Great Barrier Reef generated gales near an unprecedented number of reefs due to its unusual track while it maintained high intensity, but it was not large (Puotinen et al., 2016). We categorized the tropical cyclone track position spatial footprints that intersect with the four Australian coral reef regions into four very damaging categories (Table 1) and one category containing all others. The translation speed was calculated using the distance between consecutive track positions (length of great circle arc between coordinates of track positions) divided by the time between the recorded track positions.

2.2. Calculating Downscaled Reef Damaging Tropical Cyclone Metrics

Tropical cyclone intensity is measured by maximum wind speeds, which occur over a relatively small area at the boundary of the cyclone eye. As most cyclone eye radii in the Australian region range between 10 and 40 km, accurately identifying and assessing cyclones from a global climate model requires relatively high spatial resolution (\sim 20 km or less). This requirement combined with the fact that many tropical cyclones are short-lived, makes

The Four Reef Damaging Categories of Track Positions								
Category	Name	Description						
1	Intense and large	Maximum wind speed >33 m/s and radius to gales >275 km						
2	Intense and slow-moving	Maximum wind speed >33 m/s and translation speed <5 m/s						
3	Large and slow-moving	Radius to gales >275 km and translation speed <5 m/s						
4	Intense, large and slow-moving	Maximum wind speed >33 m/s, radius to gales >275 km and translation speed <5 m/s						

Table 1

their simulation by relatively coarse scale global climate models difficult (Knutson et al., 2010). The high spatial resolution necessary to simulate intense tropical cyclones can be obtained by dynamical and statistical-dynamical downscaling. Downscaling is thus commonly used to simulate observed tropical cyclones and examine basin-scale trends in tropical cyclone frequency and intensity (Camargo et al., 2008; Emanuel, 2006; Emanuel et al., 2008; Knutson et al., 2013, 2015; Villarini et al., 2011).

Here, we obtained downscaled tropical cyclone tracks (commonly referred to as "synthetic tracks") derived from six CMIP5 climate models (Table S2 in Supporting Information S1) from K. Emanuel (Emanuel, 2013) for the simulated historical (1985–2005), mid-century (2040–2060) and end of century (2080–2100) time periods under the RCP8.5 high-emission scenario. The tracks were downscaled using a statistical-dynamical technique. Detail on how the tracks were generated is described in Emanuel et al. (2006) and Emanuel et al. (2008). These simulations were used here due to the large number of downscaled tracks (n = 3,000) for historical, mid-century and end of century. These tracks provide robust statistics for predicting changes in tropical cyclone characteristics, especially for the less common but most destructive storms (Emanuel et al., 2006). Further, the tracks have demonstrated applicability to the Australia region (Ramsay et al., 2018) and mid-century tracks are available. Mid-century tracks are more relevant for coral reef timescales than the commonly used 2080-2100 future tracks, given the expected near-term decline of coral reefs due to climate warming (Dixon et al., 2022; Hughes et al., 2017). Finally, the downscaled tracks respond to the physics of climate change, for example, responding to changes in atmospheric water vapor, sea surface temperature and wind shear. The evolution of these large-scale, tropical cyclone-relevant parameters in CMIP5 models will therefore be reflected in the future tracks (Emanuel, 2006; Emanuel et al., 2008). Natural climate variability modes, such as El Niño Southern Oscillation (ENSO), can affect reef-damaging tropical cyclone characteristics such as intensity and duration as well as where they form and track in the Australia region (Ramsay et al., 2012). Changes to ENSO patterns under future climate change are simulated by the climate models that drive the downscaled tropical cyclone tracks and so are considered in the projected changes in tropical cyclone characteristics presented here. However, there is known uncertainty in projected changes to ENSO under future climates (Taschetto et al., 2014).

The reef damaging tropical cyclone characteristics were calculated for tracks whose spatial footprint intersects the coral reef regions for the downscaled historical, mid-century and end of century tracks. We converted the 1-min maximum wind speed provided by K. Emanuel to 10-min maximum wind speed (1-min maximum wind speed * 0.88 (Harper et al., 2010; Ramsay et al., 2018)) in order to calculate maximum intensity. The radius to gales is not explicitly simulated but can be estimated by constructing radial wind profiles using the maximum wind speed and radius to maximum wind. The historical radius to gales at each track position was calculated three times, each time constructing radial wind profiles using a different method: Holland et al. (2010), Emanuel (2010) and Emanuel and Rotunno (2011). We then compared the downscaled maximum radius to gales to the observed maximum radius to gales for each model and wind profile method using two-sided Mann-Whitney-Wilcoxon to test the null hypothesis that the observed and downscaled maximum radius to gales are from continuous distributions with equal medians. The wind profile method that produced maximum radius to gales that were not significantly different from observed at the 0.01 significance level was selected as the "best" method for each model. The selected wind profile was then used to calculate the radius to gales for the mid-century and end of century downscaled tracks. Scripts for calculating the radius to gales using the three wind profiles were provided by K. Emanuel. We calculated the duration of gales and the reef damage index in the same way as for the observed tracks and categorized the most damaging track positions in the same way as the observed tracks. Both observed and downscaled tracks were linearly interpolated to 1 hourly time steps.

2.3. Statitsical Analysis

We compared the spatial distributions of the observed and downscaled tracks at two spatial scales for the period 1985–2005: regional and reef. At the regional scale, we compared the number of tracks whose spatial footprint intersects each of the coral reef regions between simulated past and observed tropical cyclones. At the reef scale, we compared the number of tracks intersecting each of the reef areas within the coral reef regions between simulated past and observed tropical cyclones. At the reef scale, we compared the number of tracks intersecting each of the reef areas within the coral reef regions between simulated past and observed. This analysis assessed the suitability of the downscaled tracks in representing future changes to tropical cyclones for finer than regional scales (e.g., within the Great Barrier Reef). Both were compared using a Chi Squared test to test whether the distribution of observed and downscaled tracks whose spatial footprints intersect each of the regions/reefs were different. We then visually compared kernel density estimates (KDE) of the tropical cyclone hourly track positions, and the median genesis (formation) positions and track positions for the first 10 days between observed and downscaled tracks (Ramsay et al., 2018) for tropical cyclones whose spatial footprint intersects with each of the coral reef regions between 1985 and 2005.

The observed reef damaging tropical cyclone characteristics were compared to the downscaled historical metrics for each model and region separately based on a two-sided Mann-Whitney-Wilcoxon test. This analysis tested the null hypothesis that the observed and downscaled metrics for tropical cyclones intersecting each region are from continuous distributions with equal medians. *P*-values greater than 0.01 indicated that the model was not significantly different from the observed regional reef damaging tropical cyclone metrics and indicated that models capture observed cyclone characteristics. One-sided Mann-Whitney-Wilcoxon was used to test the direction of differences. Observed and downscaled maximum intensity and duration of gales were compared for the period 1985–2005. For the maximum radius to gales, we compared the longer observed 1985–2020 period to the simulated past 1985–2005 period and assumed stationarity for the periods before and after 2005. The most reliable wind radii data has been recorded in best track databases for the Australian region since 2003, with some opportunity-based surface observations included earlier in the record (Courtney et al., 2021). We also used the 1985–2020 observed period to compare the reef damage index and number of tracks in each damage category as both require reliable radius to gales data. The number of track positions in each of the damage categories was compared between the observed (1985–2020) and simulated past (1985–2005) periods using a Chi Squared test to compare the distributions of the two populations (observed and downscaled) of categorical data.

Changes in reef damaging tropical cyclone characteristics with future climate change were determined by bootstrapping the difference in means following Ramsay et al. (2018). The mean change in reef damaging metric (e.g., maximum intensity) and 95% confidence intervals were calculated from 10,000 replicates using the bias-corrected and accelerated technique. If the confidence intervals did not cross zero, the projected change was considered statistically significant. Bootstrapped changes in means were analyzed for all tropical cyclones in the Southern Hemisphere ocean basins (South Indian Ocean and South Pacific Ocean) and for tropical cyclones impacting each of the coral reef regions for both the mid-century and end of century periods.

3. Results

3.1. Comparing Observed and Downscaled Past Tropical Cyclones

The maximum radius to gales calculated using the Emanuel (2010) wind profile correctly captured past cyclone size observations for four out of six of the models (p > 0.01; GFDL, HADGEM, MIROC and MPI; Figure S1 in Supporting Information S1). For CCSM and IPSL, the maximum radius to gales was correctly represented using the Holland et al. (2010) wind profile (p > 0.01). The maximum radius to gales calculated using the Emanuel and Rotunno (2011) wind profile was significantly higher than observed for all models (p < 0.0001). For the rest of the analysis, the Emanuel (2010) wind profile was thus used to calculate the radius to gales for GFDL, HADGEM, MIROC and MPI and the Holland et al. (2010) wind profile for CCSM and IPSL for the historical and future downscaled tracks.

3.1.1. Spatial Distribution of Tracks and Cyclogenesis

The number of observed tracks whose spatial footprints intersect with each region was significantly different (p < 0.01) to the number of downscaled historical tracks for all but HADGEM ($\chi^2 = 2.39$, p = 0.495) and IPSL ($\chi^2 = 5.88$, p = 0.117). In the downscaled tracks, there was a greater proportion of tracks impacting the Coral Sea and Great Barrier Reef than observed and a smaller proportion of tracks impacting Western Australia (Figure 2).





Figure 2. Percentage of tropical cyclone tracks whose spatial footprints intersect with each coral reef region out of the total number of tracks intersecting any region in the observed (1985–2005) and simulated past (1985–2005) periods. CS—Coral Sea, GBR—Great Barrier Reef, NT—Northern Territory, WA—Western Australia.

The number of tracks impacting the reefs within the Coral Sea was captured by CCSM, GFDL, HADGEM, MIROC and MPI (i.e., there was no significant difference between the observed and downscaled past distribution of tracks). There was a greater proportion of downscaled tracks impacting the Northern reef area (1 in Figure 3a) than observed and fewer tracks impacting the central sections (2-4 in Figure 3a) than observed. The number of observed tracks whose spatial footprints intersect with reefs within the Great Barrier Reef was correctly represented by all models (p > 0.01; Table S3 in Supporting Information S1; Figure 3b). The number of observed tracks that impact reefs within Western Australia was captured by CCSM, GFDL, HADGEM, IPSL and MPI. There was a greater proportion of downscaled tracks impacting Ashmore (1 in Figure 3c) and Scott Reef (2 in Figure 3c) in the north of the region than observed. The downscaled tracks in some of the models had a smaller proportion impacting the Kimberly (3 in Figure 3c) and Montebellos (5 in Figure 3c) on the Western Australian coastline than observed. This finding highlights how uncertainty increases at finer spatial scales, making it problematic to use these data at increasingly finer scales.

The observed median track for tropical cyclones impacting Western Australia initially followed the Western Australian coastline, and therefore a single tropical cyclone had the potential to impact many reefs during its track (Figure 4). The median downscaled tracks and track positions extended further west than observed. The downscaled tracks in CCSM, GFDL, HADGEM, IPSL and MPI captured the observed median track direction traveling east to west for tropical cyclones impacting the Northern Territory, though CCSM, GFDL and IPSL had genesis positions further east than observed. The observed

median track for tropical cyclones impacting the Coral Sea and Great Barrier Reef tracked north to south while the downscaled median tracks started further east and tracked east to west before curving south. Downscaled track positions also extended further east in the north of the regions than observed. The median cyclone genesis positions in the models were at a lower latitude than observed for all coral reef regions (Figure S2 in Supporting Information S1).

3.1.2. Reef Damaging Tropical Cyclone Characteristics

The maximum intensity (Figure S3 in Supporting Information S1) was best captured for tracks whose spatial footprints intersect the Coral Sea and Great Barrier Reef, followed by Western Australia (five and four out of six models were not significantly different to observed, respectively). The maximum intensity was poorly represented for the Northern Territory (all six models were significantly different to observed). The maximum radius to gales (Figure S4 in Supporting Information S1) was captured for tracks whose spatial footprints intersect the Great Barrier Reef, Western Australia and the Coral Sea by six, five and three out of six models, respectively. All six models had significantly higher maximum radius to gales of tracks impacting the Northern Territory. The duration of gales (Figure S5 in Supporting Information S1) was best represented for tracks whose spatial footprints intersect the Great Barrier Reef and Northern Territory followed by Western Australia as six, five and two models were not significantly different to observed, respectively. None of the models captured the observed duration of gales for tracks impacting the Coral Sea. The observed reef damage index (Figure S6 in Supporting Information S1) was correctly represented for tracks whose spatial footprints intersect the Great Barrier Reef and Northern Territory. S1) was correctly represented for tracks whose spatial footprints intersect the Great Barrier Reef and Northern Territory followed by Western Australia as six, five and two models were not significantly different to observed, respectively. None of the models captured the observed duration of gales for tracks impacting the Coral Sea. The observed reef damage index (Figure S6 in Supporting Information S1) was correctly represented for tracks whose spatial footprints intersect the Great Barrier Reef and Western Australia by all six models, and the Coral Sea by four models. All six models had significantly higher reef damage indices of tracks impacting the Northern Territory.

We found that CCSM and IPSL were the best performing models, as they were not significantly different from observed on 11 and 12 out of 16 occasions, respectively (Table 2). The Great Barrier Reef was the best represented region, as downscaled metrics were not significantly different from observed on 23 out of 24 occasions, followed by Western Australia (17/24) and then the Coral Sea (12/24). The Northern Territory was poorly represented by all six models for every metric but the duration of gales (5/24).





Figure 3. Percentage of tropical cyclone tracks whose spatial footprints intersect with each coral reef within the (a) Coral Sea, (b) Great Barrier Reef, and (c) Western Australia in the observed (1985–2005) and simulated past (1985–2005) periods. The Northern Territory is not included because it is not divided into coral reef areas due to uncertainty regarding the location of coral reefs within the region. In Western Australia, the reefs are 1—Ashmore, 2—Scott Reef, 3—Kimberly, 4—Rowley Shoals, 5—Montebello, 6—Pilbara, 7—Ningaloo, 8—Shark Bay.

3.1.3. Damaging Track Positions

There were no observed or downscaled track positions in the most damaging category (intense, large and slow-moving) whose spatial footprints intersect with coral reef regions (Figure S7 in Supporting Information S1). Observations in the Coral Sea and Great Barrier Reef regions showed similar distributions across the categories, with lower percentages of intense and large cyclones compared to intense and slow and large and slow. In contrast, observations in the Northern Territory and Western Australia both showed by far the greatest percentage in the intense and slow category with relatively few in the other categories.

The downscaled tracks simulated track positions in every damage category represented by the observed tracks, indicating that the downscaled tracks are able to simulate reef damaging tropical cyclones (Figure S7 in Supporting Information S1). However, the number of downscaled historical tracks in each of the damage categories was not correctly represented by any of the models for any region (p < 0.01; Table S5 in Supporting Information S1). For the Great Barrier Reef and Coral Sea, all six models had more intense and large track positions and fewer large and slow-moving track positions than observed (Figure S7 in Supporting Information S1). For Western Australia and the Northern Territory, the models captured the low occurrence of large and slow-moving track positions. For the Northern Territory, many models had more intense and slow-moving track positions. For the Northern Territory, many models had fewer than observed of these track positions. Projections of the number of track positions in each of the damage categories in the future are not included here due to the downscaled tracks' inability to capture the observed distribution of damaging track positions.





Figure 4. Kernel density estimates (KDE) of observed (1985–2005) and downscaled past (1985–2005) tropical cyclone track positions for tropical cyclones whose spatial footprints intersect with each of the four coral reef regions. The median hourly track positions for the first 10 days are shown by the circles and the 75% KDE contours are shown by the lines. The median genesis positions are shown by the black outlined circles.

3.2. Projected Changes in Reef Damaging Tropical Cyclone Characteristics

The downscaled tracks exhibited a significant increase in the maximum intensity of tropical cyclones during the mid-century period (2040–2060) (Figure S8 in Supporting Information S1) in the CCSM, GFDL and MIROC models for the Southern Hemisphere ocean basins (South Indian Ocean and South Pacific Ocean). By the end of the century (2080–2100), five models exhibited a significant increase in maximum intensity (CCSM, GFDL, HADGEM, MIROC and MPI). The downscaled tracks in the IPSL model decreased significantly in their maximum intensity in the end of century period. Five models (CCSM, GFDL, IPSL, MIROC and MPI) exhibited a significant increase in the maximum radius to gales in the mid-century extending to all six models in the end of century (Figure S8 in Supporting Information S1). The projected change in the duration of gales showed opposite tendencies between the mid and end of century periods, with four models displaying a significant decrease (GFDL, HADGEM, IPSL and MIROC) and increase (CCSM, GFDL, HADGEM and MPI) during the mid and end of century periods, respectively. CCSM and IPSL were best able to reproduce the observed reef damaging tropical cyclone characteristics for the historical period (Table 2) but had opposite projections to each other for maximum intensity and duration of gales. Both models exhibited a significant increase in maximum radius to gales.

3.2.1. Cyclone Projections for the Mid-Century (2040–2060)

Projected changes in the reef damaging tropical cyclone characteristics varied in sign in the mid-century. The tracks driven by the MIROC model exhibited a significant increase in maximum intensity (Figure 5) for tropical cyclones impacting all four regions. The tracks in CCSM exhibited significant increases in the maximum intensity of tropical cyclones impacting the Coral Sea and Western Australia. Tracks impacting the Coral Sea, the Great Barrier Reef and the Northern Territory exhibited significant decreases in maximum intensity in a third of models (IPSL and MPI for the Great Barrier Reef and the Northern Territory, and GFDL and IPSL for the Coral Sea). For the maximum radius to gales, a significant increase was projected in four models for Western Australia and the Coral Sea, three models for the Great Barrier Reef and two models for the Northern Territory. The only model with a significant decrease in the maximum radius to gales in the maximum radius to gales was HADGEM (for tracks impacting the



Table 2

p-Values for Two-Sided Mann-Whitney-Wilcoxon Test to Test the Null Hypothesis That the Following Observed and Downscaled Metrics for Tropical Cyclones Whose Spatial Footprints Intersect Each Region Are From Continuous Distributions With Equal Medians: Maximum Intensity, Maximum Radius to Gales, Duration of Gales and Reef Damage Index

Metric	Region	CCSM	GFDL	HADGEM	IPSL	MIROC	MPI	Region performance (out of 24) ^c
Maximum intensity (m/s)	CS	0.888 ^a	0.099	0.894	0.586	0.002	0.439	5
	GBR	0.308	0.064	0.156	0.256	0.001	0.058	5
	NT	0.000	0.000	0.000	0.000	0.000	0.000	0
	WA	0.055	0.008	0.001	0.732	0.016	0.040	4
Maximum radius to gales (km)	CS	0.353	0.000	0.000	0.103	0.059	0.000	3
	GBR	0.033	0.661	0.934	0.127	0.384	0.184	6
	NT	0.000	0.000	0.000	0.000	0.000	0.000	0
	WA	0.863	0.385	0.003	0.397	0.058	0.234	5
Duration of gales (days)	CS	0.003	0.004	0.000	0.001	0.009	0.000	0
	GBR	0.346	0.356	0.313	0.226	0.638	0.345	6
	NT	0.801	0.070	0.002	0.993	0.031	0.021	5
	WA	0.000	0.087	0.004	0.015	0.001	0.007	2
Reef damage index	CS	0.193	0.081	0.003	0.087	0.690	0.003	4
	GBR	0.445	0.807	0.843	0.646	0.127	0.994	6
	NT	0.000	0.000	0.000	0.000	0.000	0.000	0
	WA	0.057	0.109	0.068	0.901	0.788	0.477	6
Regional model performance (out of 4)	CS	3	2	1	3	2	1	12
	GBR	4	4	4	4	3	4	23
	NT	1	1	0	1	1	1	5
	WA	3	3	1	4	3	3	17
Model performance (out of 16) ^b		11	10	6	12	9	9	

Note. The full results (medians, sample sizes, U test statistics and *p* values) are reported in Table S4 in Supporting Information S1.

^aBold values indicate where the test hypothesis cannot be rejected (p > 0.01); that is, the model simulates a similar distribution to observed. ^bModel performance is the number of times a model succeeds in simulating the observed distribution (p > 0.01). Model performance is out of 16 (four metrics * four regions). ^cRegion performance is out of 24 (four metrics * six models).

Northern Territory and Western Australia). For the duration of gales, only MIROC and CCSM exhibited significant increases for cyclones impacting the Coral Sea and Western Australia, respectively. A significant decrease in the duration of gales was projected for tracks impacting the Great Barrier Reef when driven by IPSL, GFDL and MPI, for the Northern Territory in the IPSL and MPI models, and for Western Australia in the GFDL and MPI models. The reef damage index exhibited mixed projections, with MIROC tracks displaying a significant increase for all four regions, and CCSM producing a significant increase for the Western Australia region only. The IPSL tracks exhibited a significant decrease in the reef damage index in three regions, while the tracks driven by HADGEM and MPI exhibited a significant decrease in two regions each.

3.2.2. Cyclone Projections for the End of the Century (2080-2100)

The MIROC model tracks exhibited a significant increase in maximum intensity during the end of century period for tropical cyclones impacting every region (Figure 6), although the projected increase was lower than for the mid-century for cyclones impacting the Coral Sea and Great Barrier Reef. Other models with significant increases in maximum intensity at the end of the century included CCSM for three regions (Coral Sea, Great Barrier Reef and Western Australia), and GFDL for two regions (Northern Territory and Western Australia). On the other hand, a significant decrease in maximum intensity was found for tropical cyclones impacting the Coral Sea, Great Sea, Great Barrier Reef and Northern Territory when driven by IPSL. For the maximum radius to gales, many models exhibited significant increases, including for tropical cyclones impacting the Coral Sea (five out of six models), the Great Barrier Reef, Northern Territory and Western Australia (four models each). The tracks driven





Figure 5. Mean projected change from the simulated past (1985–2005) to simulated mid-century (2040–2060) period in reef damaging tropical cyclone metric of 10,000 replicates. The error bars depict 95% confidence intervals.

by HADGEM were the only tracks to exhibit a significant decrease in maximum radius to gales in the end of the century, which was for tropical cyclones impacting the Great Barrier Reef and Northern Territory. For the duration of gales, the downscaled tracks exhibited a significant increase for tropical cyclones impacting the Coral Sea and Western Australia for three models each and two models each for the Great Barrier Reef. Conversely, the MPI tracks exhibited a significant decrease in the duration of gales in the Northern Territory. Finally, assessment of the reef damage index revealed that three out of six models yielded a significant increase in this metric for tropical cyclones impacting Western Australia, as did two models for the Coral Sea, Great Barrier Reef and Northern Territory. The tracks driven by MPI exhibited a significant decrease in the reef damage index for tropical cyclones impacting Western Australia.

The projected changes in coral reef damaging tropical cyclone characteristics for both the mid-century and end of century are dependent on projected changes in where tropical cyclones will track (Figures S9 and S10 in Supporting Information S1). Spatial uncertainties in downscaled track trajectories for the historical period highlight uncertainty in the projected changes to the characteristics presented here.

4. Discussion

Increased tropical cyclone peak intensities under future climate change are a well-established climate change signal at global and ocean-basin scales (Knutson et al., 2020) and are commonly cited as a threat to coral reefs (Cheal et al., 2017; França et al., 2020; Gilmour et al., 2019; Harvey et al., 2018). However, the extent to which future tropical cyclones will damage coral reefs more than in the recent historical climate has been unknown.



Figure 6. Mean projected change from the simulated past (1985–2005) to simulated end of century (2080–2100) period in reef damaging tropical cyclone metric of 10,000 replicates. The error bars depict 95% confidence intervals.

Here, we found that model projections of future reef damage in the mid century and end of century are uncertain, with some models projecting increases in future reef damage and reef damaging characteristics (e.g., intensity) and others decreases. Perhaps this is to be expected given the considerable uncertainty in the projections presented here, in part due to a limited understanding of the mechanisms that influence the reef damaging characteristics of tropical cyclones (e.g., size and translation speed) and how these will change in the future. It should also be noted that the observations themselves carry uncertainty due to the limited observational period, changes in tropical cyclone observing practices over time (Courtney et al., 2021), and the fine-scale spatial nature of the tracks examined which also limits their sample size. The similarity of tropical cyclone characteristics between the simulated and observed historical tracks was closest for the Great Barrier Reef. This finding may indicate that downscaled future tropical cyclone tracks could be credible in some instances, but nevertheless we recommend caution given the spatial uncertainties in track behavior, particularly when making predictions at the sub-regional scale, and the poor representation of the most damaging tropical cyclones for coral reefs.

4.1. Spatial Distribution of Tracks and Cyclogenesis

The formation locations and trajectories of the downscaled tracks exhibited pronounced differences in some instances compared with the observed tracks. There is therefore some uncertainty in where they will generate seas capable of damaging coral reefs. The median genesis position of downscaled tropical cyclones impacting all four coral reef regions was at lower latitudes than observed (Ramsay et al., 2018). This finding is likely due to the random seeding technique used to generate the downscaled tracks where tropical cyclones can form anywhere south of 2°S (Emanuel et al., 2008). The lower latitude of cyclogenesis positions in the downscaled

tracks influenced the subsequent track trajectories, causing more of the northern reef areas to be projected to be impacted in the past than was observed in the Coral Sea and Western Australia. The models underestimated the proportion of tropical cyclones impacting Western Australian reefs and overestimated the proportion of tropical cyclones impacting Western Australia reef. The median track for the first 10 days for tropical cyclones impacting Western Australia was further west than observed resulting in the outer reefs being more frequently impacted than observed. Differences in track direction between the models and observations may be due to misrepresentations of the large-scale steering flow which predominantly influences downscaled storm movement (Ramsay et al., 2018). The spatial uncertainties in cyclogenesis and track trajectory, shown by the median genesis and track positions, impact all subsequent metrics (intensity, size, etc.) as they determine which sections of a tropical cyclone track impacts a coral reef region. That said, there was still considerable overlap in the observed and downscaled KDE areas for both the cyclogenesis and track positions, and therefore the exposure to tropical cyclones, at the reef region scale supporting their use in comparisons of observed and downscaled tropical cyclone characteristics and projections of future change. However, the spatial uncertainty limits the suitability of future downscaled tracks for projecting changes at the scale of coral reefs within regions.

4.2. Reef Damaging Tropical Cyclone Characteristics

Only one out of six models exhibited a significant increase in the reef damage index in the mid-century, and two models in the end of century, for the Coral Sea, Great Barrier Reef and Northern Territory. For Western Australia, two and three models exhibited a significant increase in the reef damage index in the mid and end of the century periods, respectively. The significant increase was predominantly due to increases in two of the three components of the reef damage index: intensity and size. When considering the entire Southern Hemisphere region, we found that half of the models projected a significant increase in maximum intensity in the mid-century, expanding to the majority (five out of six models) by the end of the century. Previous studies reported a significant increase in the intensity of South Indian Ocean tropical cyclones (Knutson et al., 2015; Murakami et al., 2012; Yamada et al., 2017; Yoshida et al., 2017), but no significant change in the intensity of Southwest Pacific tropical cyclones in the future (Emanuel, 2021; Emanuel et al., 2008; Knutson et al., 2015; Oouchi et al., 2006; Yamada et al., 2017; Yoshida et al., 2017). Regional differences in tropical cyclone intensity projections, including the physical mechanisms behind such differences, is currently a topic of active research. This hemisphere-scale increase in intensity was not as robust when examined at coral reef scales. A significant increase in the maximum intensity was only projected by two and three out of six models in the mid-century and end of the century periods, respectively. The large increase in intensity projected by MIROC downscaled tropical cyclones may be an outlier as it was not projected by the other models, especially those that represent the observed characteristics well. This disagreement in projected changes between models highlights the importance of considering projections from a model ensemble rather than a single model. In addition, even if more of the cyclones that form in the future are more intense, the overall frequency of cyclones is most often predicted to stay the same or drop (Sobel et al., 2021). This means that the absolute frequency of intense cyclones may not rise, depending on how much overall cyclone frequency changes.

Tropical cyclone size is a key determinant of the coral reef damage extent, as demonstrated for tropical cyclone Lua in Western Australia in 2012 which caused major coral loss 800 km away from its track (Puotinen et al., 2020). Four to five out of six models projected a significant increase in the maximum radius to gales of tropical cyclones impacting all four regions by the end of century indicating a robust change in tropical cyclone size in the future. However, there is uncertainty in the mechanisms that determine tropical cyclone size and therefore how these will change in the future limiting robust projections of size at the coral reef region scale and even more so at the within reef scale. Knutson et al. (2015) and Yamada et al. (2017) found significant increases in tropical cyclone size with future warming in both the South Indian and South Pacific oceans. Kim et al. (2014) found significant increases in tropical cyclone size from a doubling of CO₂ in every ocean basin except the South Indian Ocean. Intensity is suggested to influence changes in tropical cyclone size because size metrics are often based on wind speed (e.g., radius to 17 m/s winds) (Kim et al., 2014). Here, radius to gales is estimated by constructing wind profiles based on maximum wind speed so projected changes in tropical cyclone size are influenced by changes in intensity. However, Chavas et al. (2016) find that relative sea surface temperature (i.e., surface temperature (locally) minus the tropical-mean value) is the better determinant of tropical cyclone size. Without a better understanding of these mechanisms, there will be large uncertainty in the magnitude and spatial distribution of projected coral reef damage as intensity is not the only driver of reef damage.

4.3. Damaging Track Positions

The ability of downscaled cyclone tracks to capture observed cyclone characteristics differs by geographic region. The downscaled tracks generally captured the observed characteristics of tropical cyclones that generate reef damaging wave climates (intensity, size and duration) for the Great Barrier Reef, but not the Northern Territory, and only partially for the Coral Sea and Western Australia. The distribution of downscaled tropical cyclones whose spatial footprints intersect with reef areas within the regions in most models (five to six out of six models depending on the region) was not significantly different to observed, indicating that the downscaled tracks are suitable for projecting within region changes to reef damaging tropical cyclone characteristics. However, the models do not capture the frequency of track positions in the most damaging categories which are the most important for determining severe or large-scale reef damage. Cheal et al. (2017) report that three tropical cyclones were the biggest drivers of coral decline on the Great Barrier Reef from 1985 to 2012 as shown by De'ath et al. (2012): Cyclone Hamish (2009), Cyclone Yasi (2011) and Cyclone Ita (2014). Hamish and Ita were both intense and slow-moving, and Yasi was intense and large while their spatial footprints intersected with the Great Barrier Reef. Accurately simulating these "most damaging" tropical cyclones is crucial for estimating future reef damage severity and extent. However, comparing the most reef damaging tropical cyclones between observed and downscaled historical tracks is complex because reef damaging tropical cyclones are relatively rare in the observed record. Thus, the observed past represents just one realization of possible past tropical cyclone tracks while the downscaled tracks provide a large sample of track positions. Caution is therefore recommended when interpreting projected changes to tropical cyclone-induced reef damage at the regional scale in the future in all coral reef regions.

4.4. Future Research

We focus here on physical damage to coral reefs caused by tropical cyclone-induced waves, and found considerable uncertainty in projections of the reef damage index in the mid-century (2040–2060) and the end of century (2080–2100) based on a commonly-used tropical cyclone downscaling approach (Emanuel et al., 2006). Examinations of other cyclone-related drivers of coral reef damage, such as rainfall and flooding (Van Woesik et al., 1995), should be considered alongside wave damage in future to provide a holistic view of tropical cyclone impacts to coral reefs with climate change. An alternative investigation of explicitly-simulated tropical cyclones in climate models is also recommended as recent studies have shown that such projections are sensitive to the approach used to simulate tropical cyclones (i.e., explicitly-simulated vs. downscaled) in addition to the downscaling method applied (Jing et al., 2021). Further, temporal clustering can impact coral reef degradation as successive tropical cyclone events mean that tropical cyclones following a very damaging first strike may cause relatively little further damage compared to a random regime where corals have had time to recover in between strikes (Mumby et al., 2011; Wolff et al., 2016). However, clustered tropical cyclones may also destabilize the substrate preventing the survival of coral larval recruits and inhibiting recovery (Ceccarelli et al., 2020).

Tropical cyclones interact with other anthropogenic stressors exacerbating (i.e., ocean acidification) or reducing coral reef damage. Tropical cyclones generate a cooling wake through mixing of cooler deeper waters and enhanced surface fluxes which can provide respite to coral reefs experiencing thermal stress-induced coral bleaching (Carrigan & Puotinen, 2014). Both the intensity and spatial extent of the cool wake are maximized for tropical cyclones that are both intense and slow-moving, assuming favorable ocean conditions (Mei & Pasquero, 2013). Coral reef damage risk is also dependent on the coral species present, and the depth and exposure of a site (Blackwood et al., 2011; Harmelin-Vivien, 1994). We set thresholds for damaging track positions here that are assumed have the potential to cause coral reef damage to vulnerable colonies that are present, recognizing that coral reef damage is likely to be patchy at local scales within broadly defined risk zones (Puotinen et al., 2016).

ENSO is known to impact tropical cyclone frequency, genesis, track, intensity and duration in the Southern Hemisphere causing changes in tropical cyclone activity in different parts of the region during different phases (Lin et al., 2021). For example, tropical cyclone activity in the Australia region is enhanced during La Niña years (Ramsay et al., 2012). Changes to ENSO patterns under future climate change are therefore likely to affect reef damaging tropical cyclone characteristics. However, uncertainty in how ENSO will change with climate change limits projections of the influence of ENSO on tropical cyclones in the future.



Incorporating additional factors affecting coral reef damage risk from tropical cyclones such as rainfall, temporal patterns of tropical cyclones, interacting stressors, local-scale ecological data and improved estimates of natural climate variability into future research will contribute to a holistic picture of the future tropical cyclone threat to coral reefs.

Data Availability Statement

The observed tropical cyclone tracks for the period 1985–2020 used in this analysis are publicly available online from the International Best Track Archive for Climate Stewardship (IBTrACS) at: https://www.ncei.noaa.gov/products/international-best-track-archive?name=ib-v4-access. The downscaled historical (1985–2005) and RCP8.5 mid-century (2040–2060) and end of the century (2080–2100) tropical cyclone tracks were provided by Kerry Emanuel. The tracks are to be used for nonprofit research only and so are not openly available but they can be requested for research purposes from K. Emanuel (emanuel@mit.edu). Researchers using these tracks are asked to sign a data agreement stating that the tracks will not be redistributed to ensure that the data is used only for non-profit research.

References

Beeden, R., Maynard, J., Puotinen, M., Marshall, P., Dryden, J., Goldberg, J., & Williams, G. (2015). Impacts and recovery from severe tropical cyclone Yasi on the Great Barrier Reef. *PLoS One*, 10(4), e0121272. https://doi.org/10.1371/journal.pone.0121272

- Beyer, H. L., Kennedy, E. V., Beger, M., Chen, C. A., Cinner, J. E., Darling, E. S., et al. (2018). Risk-sensitive planning for conserving coral reefs under rapid climate change. *Conservation Letters*, 11(6), e12587. https://doi.org/10.1111/conl.12587
- Blackwood, J. C., Hastings, A., & Mumby, P. J. (2011). A model-based approach to determine the long-term effects of multiple interacting stressors on coral reefs. *Ecological Applications*, 21(7), 2722–2733. https://doi.org/10.1890/10-2195.1
- Brodie, J. E., Kroon, F. J., Schaffelke, B., Wolanski, E. C., Lewis, S. E., Devlin, M. J., et al. (2012). Terrestrial pollutant runoff to the Great Barrier Reef: An update of issues, priorities and management responses. *Marine Pollution Bulletin*, 65(4–9), 81–100. https://doi.org/10.1016/j. marpolbul.2011.12.012
- Callaghan, D. P., Mumby, P. J., & Mason, M. S. (2020). Near-reef and nearshore tropical cyclone wave climate in the Great Barrier Reef with and without reef structure. *Coastal Engineering*, 157, 103652. https://doi.org/10.1016/j.coastaleng.2020.103652
- Camargo, S. J., Huilan, L., & Sun, L. (2008). Feasibility study for downscaling seasonal tropical cyclone activity using the NCEP regional spectral model. *International Journal of Climatology*, 2029(March 2008), 2011–2029. https://doi.org/10.1002/joc.1400
- Camp, J., Roberts, M. J., Comer, R. E., Wu, P., MacLachlan, C., Bett, P. E., et al. (2019). The Western Pacific subtropical high and tropical cyclone landfall: Seasonal forecasts using the Met Office GloSea5 system. *Quarterly Journal of the Royal Meteorological Society*, 145(718), 105–116. https://doi.org/10.1002/qj.3407
- Carrigan, A. D., & Puotinen, M. (2014). Tropical cyclone cooling combats region-wide coral bleaching. *Global Change Biology*, 20(5), 1604– 1613. https://doi.org/10.1111/gcb.12541
- Ceccarelli, D. M., McLeod, I. M., Bostrom-Einarsson, L., Bryan, S. E., Chartrand, K. M., Emslie, M. J., et al. (2020). Substrate stabilisation and small structures in coral restoration: State of knowledge, and considerations for management and implementation. *PLoS One*, *15*(10), e0240846. https://doi.org/10.1371/journal.pone.0240846
- Chan, J. C. L., & Gray, W. M. (1982). Tropical cyclone movement and surrounding flow relationships. *Monthly Weather Review*, 110(10), 1354–1374. https://doi.org/10.1175/1520-0493(1982)110%3C1354:TCMASF%3E2.0.CO;2
- Chand, S. S., Dowdy, A. J., Ramsay, H. A., Walsh, K. J. E., Tory, K. J., Power, S. B., et al. (2019). Review of tropical cyclones in the Australian region: Climatology, variability, predictability, and trends. Wiley Interdisciplinary Reviews: Climate Change, 10(5), e602. https://doi. org/10.1002/wcc.602
- Chavas, D. R., & Emanuel, K. A. (2010). A QuikSCAT climatology of tropical cyclone size. *Geophysical Research Letters*, 37(18), L18816. https://doi.org/10.1029/2010GL044558
- Chavas, D. R., Lin, N., Dong, W., & Lin, Y. (2016). Observed tropical cyclone size revisited. *Journal of Climate*, 29(8), 2923–2939. https://doi.org/10.1175/JCLI-D-15-0731.1
- Cheal, A. J., MacNeil, M. A., Emslie, M. J., & Sweatman, H. (2017). The threat to coral reefs from more intense cyclones under climate change. *Global Change Biology*, 23(4), 1511–1524. https://doi.org/10.1111/gcb.13593
- Chu, P. S., Kim, J. H., & Ruan Chen, Y. (2012). Have steering flows in the western North Pacific and the South China Sea changed over the last 50 years? *Geophysical Research Letters*, 39(10), L10704. https://doi.org/10.1029/2012GL051709
- Courtney, J. B., Foley, G. R., Van Burgel, J. L., Trewin, B., Burton, A. D., Callaghan, J., & Davidson, N. E. (2021). Revisions to the Australian tropical cyclone best track database. *Journal of Southern Hemisphere Earth Systems Science*, 71(2), 203–227. https://doi.org/10.1071/ES21011
- De'ath, G., Fabricius, K. E., Sweatman, H., & Puotinen, M. (2012). The 27-year decline of coral cover on the Great Barrier Reef and its causes. Proceedings of the National Academy of Sciences, 109(44), 17995–17999. https://doi.org/10.1073/pnas.1208909109
- Dietzel, A., Connolly, S. R., Hughes, T. P., & Bode, M. (2021). The spatial footprint and patchiness of large-scale disturbances on coral reefs. *Global Change Biology*, 27(19), 4825–4838. https://doi.org/10.1111/gcb.15805
- Dixon, A. M., Forster, P. M., & Beger, M. (2021). Coral conservation requires ecological climate-change vulnerability assessments. Frontiers in Ecology and the Environment, 19(4), 243–250. https://doi.org/10.1002/fee.2312
- Dixon, A. M., Forster, P. M., Heron, S. F., Stoner, A. M. K., & Beger, M. (2022). Future loss of local-scale thermal refugia in coral reef ecosystems. PLoS Climate, 1(2), e0000004. https://doi.org/10.1371/journal.pclm.0000004
- Emanuel, K. (2003). Tropical cyclones. Annual Review of Earth and Planetary Sciences, 31(1), 75-104. https://doi.org/10.1146/annurev. earth.31.100901.141259
- Emanuel, K. (2006). Climate and tropical cyclone activity: A new model downscaling approach. *Journal of Climate*, 19(19), 4797–4802. https://doi.org/10.1175/JCLI3908.1

Acknowledgments

The authors would like to thank Kerry Emanuel for providing the downscaled tracks used in this study. A. M. Dixon was supported by the UK Natural Environment Research Council (NERC) Spheres Doctoral Training Partnership (NE/ L002574/1). H. A. Ramsay acknowledges funding from the Climate Systems Hub of the Australian Government's National Environmental Science Program (NESP). We acknowledge the World Climate Research Programme's Working Group on Coupled Modeling, which is responsible for CMIP, and we thank the climate modeling groups (listed in Table S2 in Supporting Information S1 of this paper) for producing and making available their model output. For CMIP the U.S. Department of Energy's Program for Climate Model Diagnosis and Intercomparison provides coordinating support and led development of software infrastructure in partnership with the Global Organization for Earth System Science Portals.



- Emanuel, K. (2010). Tropical cyclone energetics and structure. Atmospheric Turbulence and Mesoscale Meteorology, 165–192. https://doi. org/10.1017/cbo9780511735035.010
- Emanuel, K. (2013). Downscaling CMIP5 climate models shows increased tropical cyclone activity over the 21st century. Proceedings of the National Academy of Sciences of the United States of America, 110(30), 12219–12224. https://doi.org/10.1073/pnas.1301293110
- Emanuel, K. (2021). Response of global tropical cyclone activity to increasing CO₂: Results from downscaling CMIP6 models. *Journal of Climate*, 34(1), 57–70. https://doi.org/10.1175/JCLI-D-20-0367.1
- Emanuel, K., Ravela, S., Vivant, E., & Risi, C. (2006). A statistical deterministic approach to hurricane risk assessment. Bulletin of the American Meteorological Society, 87(3), 299–314. https://doi.org/10.1175/BAMS-87-3-299
- Emanuel, K., & Rotunno, R. (2011). Self-stratification of tropical cyclone outflow. Part I: Implications for storm structure. Journal of the Atmospheric Sciences, 68(10), 2236–2249. https://doi.org/10.1175/JAS-D-10-05024.1
- Emanuel, K., Sundararajan, R., & Williams, J. (2008). Hurricanes and global warming: Results from downscaling IPCC AR4 simulations. Bulletin of the American Meteorological Society, 89(3), 347–368. https://doi.org/10.1175/BAMS-89-3-347
- França, F. M., Benkwitt, C. E., Peralta, G., Robinson, J. P. W., Graham, N. A. J., Tylianakis, J. M., et al. (2020). Climatic and local stressor interactions threaten tropical forests and coral reefs. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1794), 20190116. https://doi.org/10.1098/rstb.2019.0116
- Gilmour, J. P., Cook, K. L., Ryan, N. M., Puotinen, M. L., Green, R. H., Shedrawi, G., et al. (2019). The state of Western Australia's coral reefs. Coral Reefs, 38(4), 651–667. https://doi.org/10.1007/s00338-019-01795-8
- Groves, C. R., Game, E. T., Anderson, M. G., Cross, M., Enquist, C., Ferdaña, Z., et al. (2012). Incorporating climate change into systematic conservation planning. *Biodiversity & Conservation*, 21(7), 1651–1671. https://doi.org/10.1007/s10531-012-0269-3
- Gutmann, E. D., Rasmussen, R. M., Liu, C., Ikeda, K., Bruyere, C. L., Done, J. M., et al. (2018). Changes in hurricanes from a 13-yr convection-permitting pseudo-global warming simulation. *Journal of Climate*, 31(9), 3643–3657. https://doi.org/10.1175/JCLI-D-17-0391.1
- Harmelin-Vivien, M. L. (1994). The effects of storms and cyclones on coral reefs: A review. *Journal of Coastal Research*, (12), 211–231. Harper, B. A., Kepert, J. D., & Ginger, J. D. (2010). *Guidelines for converting between various wind averaging periods in tropical cyclone condi-*
- tions (p. 54). World Meteorological Organisation Tech, Doc(WMO/TD-1555). Harvey, B. J., Nash, K. L., Blanchard, J. L., & Edwards, D. P. (2018). Ecosystem-based management of coral reefs under climate change. *Ecology*
- and Evolution, 8(12), 6354–6368. https://doi.org/10.1002/ece3.4146
- Holland, G. J., Belanger, J. I., & Fritz, A. (2010). A revised model for radial profiles of hurricane winds. *Monthly Weather Review*, 138(12), 4393–4401. https://doi.org/10.1175/2010MWR3317.1
- Hughes, T. P. (1994). Catastrophes, phase shifts, and large-scale degradation of a Caribbean coral reef. Science, 265(5178), 1547–1551. https:// doi.org/10.1126/science.265.5178.1547
- Hughes, T. P., Barnes, M. L., Bellwood, D. R., Cinner, J. E., Cumming, G. S., Jackson, J. B. C., et al. (2017). Coral reefs in the Anthropocene. *Nature*, 546(7656), 82–90. https://doi.org/10.1038/nature22901
- Hughes, T. P., & Connell, J. H. (1999). Multiple stressors on coral reefs: A long-term perspective. *Limnology & Oceanography*, 44(3 II), 932–940. https://doi.org/10.4319/10.1999.44.3_part_2.0932
- Jing, R., Lin, N., Emanuel, K., Vecchi, G., & Knutson, T. R. (2021). A Comparison of tropical cyclone projections in a high-resolution global climate model and from downscaling by statistical and statistical-deterministic methods. *Journal of Climate*, 34(23), 9349–9364. https://doi. org/10.1175/JCLI-D-21-0071.1
- Kim, H. S., Vecchi, G. A., Knutson, T. R., Anderson, W. G., Delworth, T. L., Rosati, A., et al. (2014). Tropical cyclone simulation and response to CO₂ doubling in the GFDL CM2.5 high-resolution coupled climate model. *Journal of Climate*, 27(21), 8034–8054. https://doi.org/10.1175/ JCLI-D-13-00475.1
- Knapp, K. R., Diamond, H. J., Kossin, J. P., Kruk, M. C., & Schreck, C. J. (2018). International best track archive for climate stewardship (IBTrACS) project, version 4. ibtracs. ALL.list.v04r00.csv. https://doi.org/10.25921/82ty-9e16
- Knapp, K. R., Kruk, M. C., Levinson, D. H., Diamond, H. J., & Neumann, C. J. (2010). The international best track archive for climate stewardship (IBTrACS): Unifying tropical cyclone data. *Bulletin of the American Meteorological Society*, 91(3), 363–376. https://doi. org/10.1175/2009BAMS2755.1
- Knutson, T. R., Camargo, S. J., Chan, J. C. L., Emanuel, K., Ho, C.-H., Kossin, J., et al. (2020). Tropical cyclones and climate change assessment: Part II. Projected response to anthropogenic warming. *Bulletin of the American Meteorological Society*, 101(3), E303–E322. https://doi. org/10.1175/bams-d-18-0194.1
- Knutson, T. R., McBride, J. L., Chan, J., Emanuel, K., Holland, G., Landsea, C., et al. (2010). Tropical cyclones and climate change. Nature Geoscience, 3(3), 157–163. https://doi.org/10.1038/NGEO779
- Knutson, T. R., Sirutis, J. J., Vecchi, G. A., Garner, S., Zhao, M., Kim, H. S., et al. (2013). Dynamical downscaling projections of twenty-first-century Atlantic hurricane activity: CMIP3 and CMIP5 model-based scenarios. *Journal of Climate*, 26(17), 6591–6617. https://doi.org/10.1175/ JCLI-D-12-00539.1
- Knutson, T. R., Sirutis, J. J., Zhao, M., Tuleya, R. E., Bender, M., Vecchi, G. A., et al. (2015). Global projections of intense tropical cyclone activity for the late twenty-first century from dynamical downscaling of CMIP5/RCP4.5 scenarios. *Journal of Climate*, 28(18), 7203–7224. https://doi.org/10.1175/JCLI-D-15-0129.1
- Lin, I., Camargo, S. J., Patricola, C. M., Boucharel, J., Chand, S., Klotzbach, P., et al. (2021). ENSO and tropical cyclones. In M. J. McPhaden, A. Santoso, & W. Cai (Eds.), *El Niño southern oscillation in a changing climate, Geophysical Monograph* (Vol. 253) (First). John Wiley & Sons, Inc. https://doi.org/10.1002/9781119548164.ch17
- Madin, J. S., & Connolly, S. R. (2006). Ecological consequences of major hydrodynamic disturbances on coral reefs. Nature, 444(7118), 477–480. https://doi.org/10.1038/nature05328
- Madin, J. S., Hughes, T. P., & Connolly, S. R. (2012). Calcification, storm damage and population resilience of tabular corals under climate change. PLoS One, 7(10), e46637. https://doi.org/10.1371/journal.pone.0046637
- Maynard, J. A., Beeden, R., Puotinen, M., Johnson, J. E., Marshall, P., van Hooidonk, R., et al. (2016). Great Barrier Reef no-take areas include a range of disturbance regimes. *Conservation Letters*, 9(3), 191–199. https://doi.org/10.1111/conl.12198
- Mei, W., & Pasquero, C. (2013). Spatial and temporal characterization of sea surface temperature response to tropical cyclones. Journal of Climate, 26(11), 3745–3765. https://doi.org/10.1175/JCLI-D-12-00125.1
- Mumby, P. J., Vitolo, R., & Stephenson, D. B. (2011). Temporal clustering of tropical cyclones and its ecosystem impacts. Proceedings of the National Academy of Sciences of the United States of America, 108(43), 17626–17630. https://doi.org/10.1073/pnas.1100436108
- Murakami, H., Wang, Y., Yoshimura, H., Mizuta, R., Sugi, M., Shindo, E., et al. (2012). Future changes in tropical cyclone activity projected by the new high-resolution MRI-AGCM. *Journal of Climate*, 25(9), 3237–3260. https://doi.org/10.1175/JCLI-D-11-00415.1



- Newton, A. C., Britton, R., Davies, K., Diaz, A., Franklin, D. J., Herbert, R. J. H., et al. (2021). Operationalising the concept of ecosystem collapse for conservation practice. *Biological Conservation*, 264, 109366. https://doi.org/10.1016/j.biocon.2021.109366
- Oouchi, K., Yoshimura, J., Yoshimura, H., Mizuta, R., Kusunoki, S., & Noda, A. (2006). Tropical cyclone climatology in a global-warming climate as simulated in a 20 km-mesh global atmospheric model: Frequency and wind intensity analyses. *Journal of the Meteorological Society* of Japan, 84(2), 259–276. https://doi.org/10.2151/jmsj.84.259
- Puotinen, M. L., Drost, E., Lowe, R., Depczynski, M., Radford, B., Heyward, A., & Gilmour, J. (2020). Towards modelling the future risk of cyclone wave damage to the world's coral reefs. *Global Change Biology*, 26(8), 4302–4315. https://doi.org/10.1111/gcb.15136
- Puotinen, M. L., Maynard, J. A., Beeden, R., Radford, B., & Williams, G. J. (2016). A robust operational model for predicting where tropical cyclone waves damage coral reefs. *Scientific Reports*, 6(1), 26009. https://doi.org/10.1038/srep26009
- Ramsay, H. A., Camargo, S. J., & Kim, D. (2012). Cluster analysis of tropical cyclone tracks in the Southern Hemisphere. *Climate Dynamics*, 39(3–4), 897–917. https://doi.org/10.1007/s00382-011-1225-8
- Ramsay, H. A., Chand, S. S., & Camargo, S. J. (2018). A statistical assessment of Southern Hemisphere tropical cyclone tracks in climate models. *Journal of Climate*, 31(24), 10081–10104. https://doi.org/10.1175/JCLI-D-18-0377.1
- Sobel, A. H., Wing, A. A., Camargo, S. J., Patricola, C. M., Vecchi, G. A., Lee, C. Y., & Tippett, M. K. (2021). Tropical cyclone frequency. *Earth's Future*, 9(12), e2021EF002275. https://doi.org/10.1029/2021EF002275
- Taschetto, A. S., Gupta, A. S., Jourdain, N. C., Santoso, A., Ummenhofer, C. C., & England, M. H. (2014). Cold tongue and warm pool ENSO Events in CMIP5: Mean state and future projections. *Journal of Climate*, 27(8), 2861–2885. https://doi.org/10.1175/JCLI-D-13-00437.1
- Van Woesik, R., DeVantier, L. M., & Glazebrook, J. S. (1995). Effects of Cyclone "Joy" on nearshore coral communities of the Great Barrier Reef. Marine Ecology Progress Series, 128(1–3), 261–270. https://doi.org/10.3354/meps128261
- Villarini, G., Vecchi, G. A., Knutson, T. R., Zhao, M., & Smith, J. A. (2011). North Atlantic tropical storm frequency response to anthropogenic forcing: Projections and sources of uncertainty. *Journal of Climate*, 24(13), 3224–3238. https://doi.org/10.1175/2011JCL13853.1
- Walsh, K. J. E., Mcbride, J. L., Klotzbach, P. J., Balachandran, S., Camargo, S. J., Holland, G., et al. (2016). Tropical cyclones and climate change. Wiley Interdisciplinary Reviews: Climate Change, 7(1), 65–89. https://doi.org/10.1002/wcc.371
- Wolff, N. H., Mumby, P. J., Devlin, M., & Anthony, K. R. N. (2018). Vulnerability of the Great Barrier Reef to climate change and local pressures. *Global Change Biology*, 24(5), 1978–1991. https://doi.org/10.1111/gcb.14043
- Wolff, N. H., Wong, A., Vitolo, R., Stolberg, K., Anthony, K. R. N., & Mumby, P. J. (2016). Temporal clustering of tropical cyclones on the Great Barrier Reef and its ecological importance. Coral Reefs, 35(2), 613–623. https://doi.org/10.1007/s00338-016-1400-9
- Wu, L., Chou, C., Chen, C. T., Huang, R., Knutson, T. R., Sirutis, J. J., et al. (2014). Simulations of the present and late-twenty-first-century western North Pacific tropical cyclone activity using a regional model. *Journal of Climate*, 27(9), 3405–3424. https://doi.org/10.1175/ JCLI-D-12-00830.1
- Yamada, Y., Satoh, M., Sugi, M., Kodama, C., Noda, A. T., Nakano, M., & Nasuno, T. (2017). Response of tropical cyclone activity and structure to global warming in a high-resolution global nonhydrostatic model. *Journal of Climate*, 30(23), 9703–9724. https://doi.org/10.1175/ JCLI-D-17-0068.1
- Yoshida, K., Sugi, M., Mizuta, R., Murakami, H., & Ishii, M. (2017). Future changes in tropical cyclone activity in high-resolution large-ensemble simulations. *Geophysical Research Letters*, 44(19), 9910–9917. https://doi.org/10.1002/2017GL075058
- Zinke, J., Gilmour, J. P., Fisher, R., Puotinen, M., Maina, J., Darling, E., et al. (2018). Gradients of disturbance and environmental conditions shape coral community structure for south-eastern Indian Ocean reefs. *Diversity and Distributions*, 24(5), 605–620. https://doi.org/10.1111/ ddi.12714

References From the Supporting Information

- Castro-Sanguino, C., Bozec, Y. M., Callaghan, D., Vercelloni, J., Rodriguez-Ramirez, A., Lopez-Marcano, S., et al. (2022). Coral composition and bottom-wave metrics improve understanding of the patchiness of cyclone damage on reefs. *Science of the Total Environment*, 804, 150178. https://doi.org/10.1016/j.scitotenv.2021.150178
- Done, T. J. (1992). Effects of tropical cyclone waves on ecological and geomorphological structures on the Great Barrier Reef. Continental Shelf Research, 12(7–8), 859–872. https://doi.org/10.1016/0278-4343(92)90048-O
- Fabricius, K. E., De'ath, G., Puotinen, M. L., Done, T., Cooper, T. F., & Burgess, S. C. (2008). Disturbance gradients on inshore and offshore coral reefs caused by a severe tropical cyclone. *Limnology & Oceanography*, 53(2), 690–704. https://doi.org/10.4319/lo.2008.53.2.0690
- McLeod, I. M., Williamson, D. H., Taylor, S., Srinivasan, M., Read, M., Boxer, C., et al. (2019). Bommies away! Logistics and early effects of repositioning 400 tonnes of displaced coral colonies following cyclone impacts on the Great Barrier Reef. *Ecological Management and Resto*ration, 20(3), 262–265. https://doi.org/10.1111/emr.12381